Author's response to reviews

Title: Assessing and Reporting Heterogeneity in Treatment Effects in Clinical Trials: A Proposal

Authors:

David Kent (dkent1@tuftsmedicalcenter.org)
Peter Rothwell (peter.rothwell@clneuro.ox.ac.uk)
Doug Altman (doug.altman@cs.ox)
Rodney Hayward (rod.hayward@med.va.gov)
John Ioannidis (jioannid@cc.uoi.gr)

Version: 2 Date: 1 July 2010

Author's response to reviews: see over
July 1, 2010

Dear Dr. Schulz,

Please find attached a revision of our submission, “Assessing and Reporting Heterogeneity in Treatment Effects in Clinical Trials: A Proposal.” We would like to thank the reviewers for their careful reading and thoughtful comments and criticisms. We feel their input has improved the manuscript and we hope that you find it acceptable for publication in Trials.

Sincerely,

David M Kent, MD, MS
Associate Professor of Medicine
Institute for Clinical Research and Health Policy Studies
Tufts Medical Center
800 Washington St, Box 63
Boston, MA, 02111